Author's response to reviews

Title: Effects of ibuprofen on the physiology and outcome of rabbit endotoxic shock

Authors:

Dr Ilhami Celik (ilhamicelik@hotmail.com)
Prof Ayhan Akbulut (a_akbulut@yahoo.com)
Sirri S Kilic (sskilic@yahoo.com)
Ali Rahman (arahman33@hotmail.com)
Pervin Vural (pervinvural@dostmail.com)
Mukaddes Canbaz (mcanbaz@istanbul.edu)
Suleyman Felek (feleks@hotmail.com)

Version: 7 Date: 31 Oct 2002

1- Title: I could not find any unnecessary capitalisation from the title.
2- References: I arranged them in square brackets.
3- Competing interest was written.
4- Figures: They are cropped.